Author’s response to reviews

Title: Polymorphisms in the epidermal growth factor receptor gene and the risk of primary lung cancer: a case-control study

Authors:

Jin Eun Choi (ddochi74@hanmail.net)
Sun Ha Park (sunhapark@hanmail.net)
Kyung Mee Kim (96kmkim@hanmail.net)
Won Kee Lee (wonlee@knu.ac.kr)
Sin Kam (kamshin@knu.ac.kr)
Sung Ick Cha (sicha@knu.ac.kr)
Chang Ho Kim (kimch@knu.ac.kr)
Young Mo Kang (ymkang@knu.ac.kr)
Young-Chul Kim (kyc0923@chonnam.ac.kr)
Sung Beom Han (sbhan@dsmc.or.kr)
Tae Hoon Jung (thjung@knu.ac.kr)
Jae Yong Park (jaeyong@knu.ac.kr)

Version: 3 Date: 20 October 2007

Author’s response to reviews:

October 22 14, 2007

Editor-in-Chief

BMC Cancer

Dear Editor

It is my great pleasure to get a letter saying that our manuscript MS-1491149837146747, entitled “Polymorphisms in the epidermal growth factor receptor gene and the risk of primary lung cancer: a case-control study”, is accepted in BMC Cancer. We have revised our manuscript according the requests of the publisher.

Please do not hesitate to contact us if you have any questions regarding this manuscript. We look forward to your reply.

Very sincerely yours,

Jae Yong Park, M.D., Ph.D.

Department of Internal Medicine, School of Medicine

Kyungpook National University, Samduk 2Ga 50, Daegu, 700-412, Korea

Phone: 82-53-420-5536

Fax: 82-53-426-2046
E-mail: jaeyong@kyungpook.ac.kr